Author's response to reviews

Title: Low levels of vitamin C in dialysis patients is associated with decreased prealbumin and increased C-reactive protein

Authors:

Kunying Zhang (zhangky6159@163.com)
Li Liu (liulimd@gmail.com)
Xuyang Cheng (chengxuyang@medmail.com.cn)
Jie Dong (dongjie@medmail.com.cn)
Qiuming Geng (gengqm@yahoo.com.cn)
Li Zuo (zuolimd@gmail.com)

Version: 4 Date: 24 April 2011

Author's response to reviews: see over
Dear Editor(s):

Thank you for publishing our work “Low levels of vitamin C in dialysis patients is associated with decreased prealbumin and increased C-reactive protein” in this important journal. Please kindly find our response to reviewers as follows:

(1) References: Please provide full reference details for reference 34.
   To this question, we have provided full reference details for reference 34 in the part of reference.

(2) Tracking: Please remove all forms of tracking present on the manuscript.
   To this question, we have removed all forms of tracking present on the manuscript.

(3) In addition, we also corrected some typing errors in the manuscript.

Sincerely yours,

Li Zuo M.D.
Institute of Nephrology
Peking University First Hospital
8 Xishiku Street, Xicheng District
Beijing, 100034
Tel: +86(10)8357-5740
Fax:+86(10)6655-1055
Email: ZuoLiMD@Gmail.com